Author's response to reviews

Title: Semi-automated segmentation and quantification of adipose tissue in calf and thigh by MRI: a preliminary study in patients with monogenic metabolic syndrome

Authors:

Salam A Al-Attar (salattar@uwo.ca)
Rebecca L Pollex (rpollex@robarts.ca)
John F Robinson (robinson@robarts.ca)
Brooke A Miskie (bmiskie@robarts.ca)
Rhonda Walcarius (rhonda.walcarius@swri.ca)
Brian K Rutt (brian.rutt@imaging.robarts.ca)
Robert A Hegele (hegele@robarts.ca)

Version: 4 Date: 30 August 2006

Author's response to reviews: see over
August 30th, 2006

Dear members of the BioMed Central Editorial Production Team,

Thank you for your response and suggestions.

We have made the suggested changes as listed below:

- Affiliations - authors have been linked to their corresponding affiliation with the use of superscript numbers
- Affiliations - All affiliation were re-written in full following the format suggested (department, institute, city, and country).
- The word count was removed from the manuscript
- ImageJ’s website address was cited in the reference list and removed from the manuscript text (see Reference 9).
- Figures - We have provided larger, more resolved images of Figures 2 and 3. Figure 1 is already a large size with good resolution.

Please let us know if we can be of further assistance.

Sincerely,

Salam A. Al-Attar,
BSc- Honours Genetics Graduate
MSc- Biochemistry Candidate

and

Robert A. Hegele, MD FRCPC FACP
406-100 Perth Drive
London, Ontario, Canada N6A 5K8
Tel: 519.663.3461; Fax: 519.663.3037; Email: hegele@robarts.ca